Dear Editor,

Thank you for accepting our article for publication. We are now ready to submit the final version of our manuscript called "Heart failure diagnosis in primary health care: clinical characteristics of problematic patients. A Clinical Judgement Analysis study".

Since we found no further suggestions for improvements in the reviewers' reports, we have only made one change in the text: On page 12, line 4 from the bottom, "values of variables" has been changed to "variable values". We have also cropped the figures as much as we think would be possible.

Yours Sincerely

Ylva Skaner

PS: I will be out of reach for all internet and e-mail correspondance between the 20th of September and the 4th of October, so if you try to contact me during this period and don't get an answer, it is not because I am rude... YS